Author’s response to reviews

Title: Torsades de pointes during laparoscopic adrenalectomy of a pheochromocytoma: a case report

Authors:

Kinge K van der Heide (k.van.der.heide@cso.umcg.nl)
Ann A de Haes (anndehaes@gmail.com)
Götz JKG Wietasch (J.K.G.Wietasch@anest.umcg.nl)
Ans ACP Wiesfeld (A.C.P.Wiesfeld@thorax.umcg.nl)
Herman HGD Hendriks (H.G.D.Hendriks@anest.umcg.nl)

Version: 3 Date: 8 March 2011

Author’s response to reviews:

Comments to authors:

JMCRep
- It is relevant, but not commented in the Discussion, that preoperative magnesemia was not tested preoperatively and resulted into abnormally low values when tested intraoperatively. Furthermore the combonation with flecainide could be avoided.
  # stressed on page 9
- The Authors mention that preoperatively testing for genetic long QT interval was not accomplished., but is not clear if it would have changed the clinical course of this case. Is it really indicated/mandatory?
  # we meant POSToperative testing for genetic long QT, so I changed it in the text (p8), since it wasn’t clear enough considering your comment.
- In patients at risk for perioperative arrhythmias preoperative electrolytes should be –if possible- normalized and concomitant medications that can favour acute arrhythmias should be minimized.
  # stressed on page 9/10

Quality of written English: we asked a native English speaker to oversee our manuscript.